Title: Full Accuracy of Machine Learning for Differentiation between Optic Neuropathies and Pseudopapilledema

Authors:
Jin Mo Ahn (jmoa9203@gmail.com)
Sangsoo Kim (sskimb@gmail.com)
Kwang-Sung Ahn (kwangsung.ahn@gmail.com)
Sung-Hoon Cho (sunghoon.cho88@gmail.com)
Ungsoo Kim (ungsookim@kimeye.com)

Version: 4 Date: 24 Jul 2019

Author’s response to reviews:

1 - We note that the current submission contains textual overlap with other previously published works. Please re-write these sentences and phrases in your own words to ensure no overlap. If there is overlap in the Methods section, please ensure that you summarize the methods and cite the source.

There is considerable overlap with:
https://journals.plos.org/plosone/article?id=10.1371%2Fjournal.pone.0207982

We recommend the authors use an anti-plagiarism software, such as turnitin or other freely available ones, to assess the overlap, and reduce it.

--> We revised the manuscript.

2 - Please cite references in the text and in the References section as recommended in our instructions to authors.
--> Instead of citing reference, we revised manuscript

3 - At this stage, please upload your proofread manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethrough or text in different colours. All relevant tables and figures should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Should you wish to respond to these revision requests, please include the
information in the designated input box only.

--> Clean version is uploaded.